Author’s response to reviews

Title: Diagnostic value of retrospective PET-MRI fusion in head-and-neck cancer

Authors:

Denys J. Loeffelbein (loeffelbein@mkg.med.tum.de)
Michael Souvatzoglou (msouvatz@yahoo.de)
Veronika Wankerl (Veronika.Wankerl@gmx.de)
Julia Dinges (Julia.Dinges@gmx.de)
Lucas M. Ritschl (ritschl@mkg.med.tum.de)
Thomas Mücke (th.mucke@gmx.de)
Anja Pickhard (a.pickhard@lrz.tu-muenchen.de)
Matthias Eiber (matthias.eiber@tum.de)
Markus Schwaiger (markus.schwaiger@tum.de)
Ambros Beer (ambros.beer@tum.de)

Version: 3 Date: 10 October 2014

Author’s response to reviews:

Dafne Solera
Executive Editor
BioMed Central
10/10/2014

MS: 2559740781204083, 2nd revision
Diagnostic value of retrospective PET-MRI Fusion in Head-and-Neck Cancer

Dear Professor Solera
Dear Ms. Battad
dear associate Editors

Thank you again for reviewing our manuscript.
Please find enclosed our 2nd revision of our manuscript for consideration as an original article for the Journal of BMC Cancer.

We included a detailed ethics statement in the methods section with an approval number of the ethical committee of the medical faculty of the Technische Universität München (approval number 366/14).

We hope that we have suitably attended to the reviewers’ comments and the editorial request. We look forward to hearing from you after the reviewing process and will be delighted if the manuscript is accepted for publication in your journal.

If you have any further questions, please do not hesitate to contact us.
Yours sincerely,

Priv.-Doz. Dr. Dr. Denys J. Loeffelbein